Author’s response to reviews

Title: Structure and governance of health and welfare systems relevant to diabetes self-management: a realist review to facilitate a European Project implemented in six countries

Authors:

Antonis Kousoulis (antonis.kousoulis@gmail.com)
Evridiki Patelarou (evridiki.patelarou@kcl.ac.uk)
Sue Shea (sueshea1@otenet.gr)
Christina Foss (i.c.foss@medisin.uio.no)
Ingrid A Ruud Knutsen (i.a.r.knutsen@medisin.uio.no)
Elka Todorova (elka.todorova@me.com)
Poli Roukova (poliroukova@yahoo.com)
MC Portillo (mportillo@unav.es)
María J Pumar-Méndez (mpumarme@unav.es)
A Mujika (amujika@unav.es)
Anne Rogers (A.E.Rogers@soton.ac.uk)
Vassilev Ivaylo (I.I.Vassilev@soton.ac.uk)
Manuel Serrano-Gil (manuel.serrano@globalalliancesms.org)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 5
Date: 1 September 2014

Author’s response to reviews: see over
Dear Editor,

On behalf of all the co-authors, thank you for the opportunity to publish our manuscript in your prestigious journal. In response to your latest comment:

1. The ethics statement has been removed.

2. We have completed and submit the PRISMA checklist. It should be noted that the paper is a Realist Review, which while draws on the principles of Systematic Reviews, still has a few differences that may not make the PRISMA checklist an exact match for its reporting. Please, note that Figure 1 is a flow chart, closely (but not exactly due to the type of the study) styled to the PRISMA flow chart.

3. The attached latest version of the manuscript has been reviewed by a native English speaker.

4. The Authors' Contributions statement has been revised as indicated.

On behalf of all the co-authors, we are looking forward for the publication.

My kindest regards,

Christos Lionis

Professor of General Practice and Primary Care
Head of the Clinic of Social and Family Medicine
School of Medicine
University of Crete
Greece
Tel: +30 2810 394621
Fax: +30 2810 394606